Author's response to reviews

Title: Successful Staged Repair for a Rare Type of Truncus Arteriosus With Interruption of the Aortic Arch and Abnormal Origin of the Left Coronary Artery

Authors:

Shunji Uchita (suchita@wakayama-med.ac.jp)
Yorikazu Harada (harady@naganoch.gr.jp)
Kentaro Honda (honda-k@wakayama-med.ac.jp)
Koji Toguchi (gucchanz19440331@yahoo.com)
Yoshiharu Nishimura (nishim-y@wakayama-med.ac.jp)
Tomohiro Suenaga (tsuenaga@wakayama-med.ac.jp)
Takeshi Takeuchi (ttake@wakayama-med.ac.jp)
Hiroyuki Suzuki (hsuzuki@wakayama-med.ac.jp)
Yoshitaka Okamura (y-ok@wakayama-med.ac.jp)

Version: 2  Date: 1 May 2013

Author's response to reviews: see over
May 1, 2013

Prof. David Taggart  
Editor-in-Chief  
Journal of Cardiothoracic Surgery

Dear Prof. Taggart,

Thank you for your Editorial Assistant’s e-mail of April 10, 2013, for our manuscript entitled “Successful Staged Repair for a Rare Type of Truncus Arteriosus With Interruption of the Aorta and Abnormal Origin of the Left Coronary Artery”

We are again submitting our revised manuscript.

Points of the revision and responses to reviewers’ suggestions are summarized in the separate sheets.

We hope that the revised manuscript is now suitable for publication of “Journal of Cardiothoracic Surgery”.

Thank you for considering our manuscript.

Sincerely yours,

Shunji Uchita, MD, PhD

Wakayama Medical University, Department of Thoracic and Cardiovascular Surgery  
811-1 Kimiidera, Wakayama-city, Wakayama 641-8510, Japan  
(+81) 73-447-0651  
(+81) 73-446-4761  
suchita@wakayama-med.ac.jp